

Contents lists available at ScienceDirect

Energy Research & Social Science



journal homepage: www.elsevier.com/locate/erss

Why does the European Union produce biofuels? Examining consistency and plausibility in prevailing narratives with quantitative storytelling



Juan J. Cadillo-Benalcazar^a, Sandra G.F. Bukkens^a, Maddalena Ripa^a, Mario Giampietro^{a,b,*}

^a Institut de Ciència i Tecnologia Ambientals, Universitat Autònoma de Barcelona, Bellaterra, Spain ^b Institució Catalana de Recerca i Estudis Avançats (ICREA), Barcelona, Spain

ARTICLE INFO

Keywords: Biofuels Renewable Energy Directive European Union Narratives Targets Quantitative storytelling The Netherlands Used cooking oil

ABSTRACT

This paper seeks to clarify the confusion created by the simultaneous use of non-equivalent policy discourses about biofuels within the EU and addresses the inconsistency between long-term goals and short-term targets. To this purpose, a novel approach, quantitative storytelling, is employed to examine the plausibility of current policy narratives. It confronts quantitative data on the production and use of biofuels at member-state (the Netherlands) and EU level against the official storytelling. Our analysis shows that in the Netherlands the actual production and use of biofuels are motivated by economic incentives and trade opportunities rather than the original policy justifications and hitting biofuel targets has become a justification in itself (the phenomenon of displacement). At EU level, the vast majority of the current (inadequate) supply of liquid biofuels still consists in first-generation biofuels, which are to be phased out. Advanced and food-waste-based biofuels are unlikely to fill the void any time soon. The analysis shows incoherence among the narratives underlying the justification of policies and those used for their implementation. It is concluded that the quality of the biofuels policy debate could be improved by acknowledging the existence and influence of untold stories and hidden justification narratives.

1. Introduction

The complex trajectory of EU biofuels policy has attracted considerable attention in the scientific literature [1–6]. This is not surprising given its turbulent history, its challenging governance [7,8], and controversies regarding its impact [9]. The very idea of replacing nonrenewable petroleum products with biofuels produced from agriculture dates back to at least the 1940s (to the use of biofuels in internal combustion engines in the USA) [10,11]. So does the criticism. "It is said that we should use alcohol and vegetable oils after the petroleum energy has been exhausted. This reminds one of Marie Antoinette's advice to the Paris poor to eat cake when they had no bread. Alcohol and vegetable oils are, of course, more expensive than petroleum fuels. Moreover, if all our crops were converted to alcohol and oil, they would not supply the present rate of petroleum-energy consumption"-Brody, 1945, p. 968 [12]. Detailed scientific assessments proving the inefficiency and high environmental impact of biofuels-and biomass in general-date back to the early 1980s [13–15]. MacKay, a highly distinguished British scholar, in his 2009 "Sustainable Energy-Without the Hot Air", argued that biomass is so inefficient we ought not even being talking about it

[16]. Yet in 2020, we still are, policy-makers and scientists alike.

The persistent presence of biofuels on the EU's sustainability agenda is part of a more generalized problem that is generated by the powerful attraction of sociotechnical imaginaries [17,18]. Jasanoff and Kim [19] have defined socio-technical imaginaries as the production of collective visions of good and attainable futures through the advancement of science and technology. Indeed, the stability of the EU increasingly depends on rosy visions portraying painless solutions to sustainability problems thus avoiding uncomfortable discussions about our current life style and standard of living [20,21]. Official storytelling about energy policy is intoxicated by 'hero stories' (as opposed to learning or caring stories) [22]. As a result, policy discussions on alternative energy sources have become muddled, mixing justifications, strategies and explanations with expectations and desired solutions.

In this paper, we investigate the co-existence of different narratives surrounding the long-term goals and short-term targets for biofuels and the related (often contrasting) "scientific evidence". The analysis is motivated by the central question: Why does or should the EU produce biofuels? To answer this question, the following issues are examined:

* Corresponding author. *E-mail address:* mario.giampietro@uab.cat (M. Giampietro).

https://doi.org/10.1016/j.erss.2020.101810

Received 8 April 2020; Received in revised form 21 September 2020; Accepted 26 September 2020

2214-6296/© 2020 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

- The policy discourses at EU level about the role that biofuels can and should play in our quest for sustainability (some of which have been officially endorsed in the formulation of policies, while others have implicitly influenced the political decision-making process);
- 2. The quality of the existing narratives about biofuels in relation to what is actually happening with biofuel production and consumption at member state and EU level, using a novel approach called Quantitative Story-Telling and using the Netherlands as a case study;
- 3. The consistency between justification, normative and explanation narratives in this policy domain.

Justification narratives are here defined as the concerns to be addressed, i.e., the overall goals to pursue. Normative narratives are the strategies and tactics adopted toward achieving these goals (actions to be taken), i.e., the provisions in the directives. Explanation narratives refer to the mechanism through which the strategies and tactics are expected to achieve the goals, i.e., the 'scientific evidence' used in the process of decision-making. These definitions are based on the report of Felt and Wynne [23].

The paper is structured as follows. Section 2 summarizes the terminology surrounding biofuels. Section 3 presents an overview of the various EU Directives on biofuels, i.e., the 'official' storytelling in the past 20 years. Section 4 presents the methodology adopted—quantitative storytelling—for performing a consistency check on EU biofuels policy in relation to the performance of the biofuel system. Section 5 presents the results of this check in relation to what is actually going on in the Netherlands and in the EU in terms of production and consumption of biofuels. The case study on the Netherlands focuses on the phenomenon of displacement (i.e., the target becoming a justification in itself) while the findings at EU level are discussed in relation to the discordance among the official justification, normative, and explanation narratives shaping EU biofuel policies. Section 6 concludes.

2. What are biofuels? - the Tower of Babel

One possible explanation for the confusion surrounding the biofuels debate lies in the confusion around the term biofuel itself and its categorization. What is a biofuel? While the definition of biofuels is legalized in the EU Directive on the Promotion of the Use of Energy from Renewable Sources (hereafter referred to as RED) -----biofuels' means liquid fuel for transport produced from biomass" [24] (p. 104), the term is commonly used to refer to other types of energy carriers based on bioenergy (e.g., electricity and heat from firewood, woodchips, biomass pellets, biogas) in legal interpretations outside the EU, in particular in the USA, as well as in scientific and grey literature [25-27]. The categorization of biofuels has gradually changed in time, growing in complexity with the introduction of additional sustainability and greenhouse gas emissions saving criteria in the various updates of the RED [24,28]. At present, categorization is based on the type of feedstock (primary energy sources) used for its production, the ecological impact and potential alternative uses of the land from which the feedstock is obtained, and its climate mitigation potential. The criteria used in the categorization are relevant for the feasibility of meeting the biofuel targets set and for qualifying for financial incentives.

The European Commission uses the following definitions [26,29]: Conventionally produced or first-generation biofuels are "biofuels produced from food crops, such as sugar, starch and vegetable oils. They are produced from land using feedstock which can also be used for food and feed". Advanced or second and third generation biofuels are defined as "biofuels produced from feedstock that do not compete directly with food and feed crops, such as wastes and agricultural residues (i.e. wheat straw, municipal waste), non-food crops (i.e. miscanthus and short rotation coppice) and algae." Note, however, that with the 2018 recast of the Renewable Energy Directive, biofuels produced from used cooking oil and animal fats are explicitly excluded from the definition of advanced biofuels ("advanced biofuels' means biofuels that are produced from the feedstock listed in Part A of Annex IX", p. 104 of [24]) and are loosely referred to as 'other biofuels' (thus formalizing the distinction already introduced in the 2015 amendment, Directive 2015/1513, [28]). Article 29 of the 2018 recast of the RED defines minimum GHG savings requirements for biofuels destined for the transport sector (compared to fossil fuels), which gradually become stricter in time [24].

Note that terminology in the USA differs markedly. The revised Renewable Fuel Standard (RFS2), based on the Energy Independence and Security Act (EISA) of 2007 [30], distinguishes between two renewable fuel categories: conventional biofuel and advanced biofuel. Conventional biofuel is defined as 'renewable fuel that is ethanol derived from corn starch'. Advanced biofuels comprise cellulosic biofuel, biomass-based diesel, and other advanced biofuels that meet life cycle GHG emissions thresholds, specific for each category, requiring a percentage improvement relative to the emissions baseline of the gasoline and diesel they replace. The latter category of 'other advanced biofuels' includes, among others, ethanol derived from sugar or starch (other than corn starch) and ethanol derived from waste material (including animal waste and food waste). Note that the RFS2 designates Brazilian sugarcane ethanol as an advanced biofuel with high mitigation potential (50%+ reduction in GHG emissions) [31,32].

The heterogeneous and changing classification of biofuels and the arbitrarily selected methods and assumptions used to measure compliance with the sustainability criteria (notably GHG emissions, e.g. [33–35])—as well as the changing regulatory frameworks and moving targets—have created a Tower of Babel situation in which the confusion distracts from the bigger picture (why are we producing biofuels in the first place?), focusing the attention on the targets and the criteria associated with incentives instead, outside of a proper semantic context.

For sake of clarity, throughout this paper, we use the legal definitions and classification of the European Commission, as laid down in the recast of the Renewable Energy Directive [24].

3. EU directives on biofuels – more than two decades of storytelling

The promotion of biofuels in the EU started back in the 1990s and has been implemented through a variety of policy instruments dealing with distinct aspects of the biofuels production chain. The evolving Common Agricultural Policy (CAP) supported the growth of energy crops, the Fuel Quality Directives allowed blending, and the Energy Taxation Directive made it possible for Member States to grant tax reductions/exemptions in favor of biofuels [36,37]. We provide here a short chronological overview, with an emphasis on the motivations (justification narratives) used for promoting biofuels on the EU political agenda.

The first motivation for the political debate on biofuels was related to agriculture [3,36,38,39]. The 1980s saw the problem of surplus wine and grain production in the EU. The crisis distillation mechanism of the wine Common Market Organisation created the need for an outlet of bioethanol produced from wine surplus [40]. The 1992 CAP reform required farmers to set aside 10 per cent of land mainly to counteract the surpluses of cereals. In 1993, a Non-Food Set Aside scheme (NFSA) was introduced as part of the CAP that allowed set aside land to be planted with energy crops, while farmers could still claim the set-aside premium [41]. The CAP Reform of 2003 introduced an additional aid mechanism supporting the production of energy crops also on non-set-aside land (decoupling) through the Energy Crop Scheme, which amounted to \notin 45 per hectare with a maximum guaranteed area of 1.5 million hectares as the budgetary ceiling [42].

Even if around the turn of the century two other justification narratives—reducing global greenhouse gas emissions and guaranteeing security of energy supply—enter into the official discourse (e.g., [43–45]), it is commonly assumed that the 2003 Biofuel Directive, the first EU directive setting (indicative) targets for biofuels in transport, was born from the political ambition to develop new market outlets for the energy crops promoted through the CAP [37,38]. Agricultural interests remained relevant in shaping policy decisions about biofuels and the support for bioenergy eventually eliminated from the first pillar of the CAP in 2008 and 2010 reentered in the second pillar—rural development policy [46,47]—through several measures supporting bioenergy development, thus adding a further justification narrative to the discourse over biofuels—i.e., sustainable rural development [2,37].

The EU Biofuel Directive (Directive 2003/30/EC, 8 May 2003) formally aimed at contributing to reducing CO₂ emissions from transport, reducing dependence on imported energy (security of energy supply), and creating new opportunities for sustainable rural development [48]. The directive required Member States to strive for the replacement of at least 5.75% of transport fossil fuels with biofuels by 2010, with an intermediate target of 2% by the end of 2005 (see Table 1). Note that the targets were not mandatory and calculated on the basis of the energy content of all petrol and diesel for transport purposes placed on their markets. Member States could deviate from the reference values in the Directive when justified.

In 2009, following concerns of the Commission about Member States failing to meet their indicative targets [49] and the publication of a Biomass Action Plan [50] and comprehensive EU Strategy for Biofuels [51], the Biofuel Directive of 2003 was repealed by the Renewable Energy Directive (2009/28/EC) (RED-I). This directive sets a binding 10% target for renewables in transport for 2020, and introduces the multiplecounting of renewable electricity and advanced biofuels [52] (see Table 1). Although the 10% target includes not only biofuels, but also other renewable energy sources, the Commission (still) considers that "biofuels are the only available large scale substitute for petrol and diesel in transport" [53]. That same year also saw an amendment of the Fuel Quality Directive (FQD) imposing that the road transport fuel mix in the EU should be 6% less carbon intensive than the fossil diesel and gasoline baseline by 2020 [54], thus complementing RED-I. Reducing greenhouse gas emissions from transport and securing energy supply through reduced dependence on (oil) imports and diversification of energy sources remain the dominant motivations [51].

Meanwhile, the food price crisis of 2007–2008 had ignited a heated debate on the use of food as fuel feeding the cars of the rich [55–59] and

evidence that crop-based fuels cannot provide significant GHG savings, if any [60–62], nor fossil fuel savings [63,64] was rapidly accumulating. The growing concern for the impact of indirect land-use changes (ILUC) on GHG emissions provided an important blow to the plausibility of the justification narrative associating the production of biofuels to a reduction of emissions [60,65-67]. In 2012, acknowledging the evidence that biofuels not necessarily contribute to the desired reduction of GHG emissions due to the effect of (indirect) land-use changes, as well as the evidence of other detrimental effects of biofuels production-such as the negative impacts on both food security [68] and the environment through destruction of tropical forests, loss of biodiversity, depletion of water resources, pollution of water due to the use of fertilizers [57,69]the European Commission proposed a cap of 5% on the contribution of conventionally produced (first-generation) biofuels to the EU's 2020 transport mix [70]. However, this proposal was not well received by the Council and the Parliament and some of the Member States pressed the Commission to raise the cap to 7% and to ignore the ILUC-effects in the accounting of greenhouse gas emissions [71].

A communication from the European Biodiesel Board accused the Commission of "purposely causing the death of the whole EU biodiesel industrial sector" [72] as the ILUC proposal would result in "closing hundreds of production sites worth many billions euros of recent investments and driving to the immediate loss of 50,000 direct and 400,000 indirect employments in the EU biodiesel production chain" [72]. Note that Charles et al. [73,74] estimated that in 2011 alone, the EU allotted approximately 5.5–6.9 billion EUR to subsidizing biofuels. This represents a significant amount if one considers that the size of the EU biofuel market was estimated at around 13–16 billion EUR for that year.

The RED-I was eventually amended by the ILUC directive in 2015 (Directive 2015/1513) [75], limiting the maximum contribution of biofuels to 7% of transport energy and requiring reporting of their ILUC emissions. Nonetheless, these emissions are not accounted for in the evaluation of the GHG performance of a biofuel, thus leaving open the possibility for high-emitting biofuels to access the Common Market and receive support [76]. As observed by Purkus et al. [4], the Council 2017

Table 1

Summary of the main targets for the use of renewable energy in the transport sector (mainly road and rail) by directives. Of these directives, all but the TBD are currently in force (simultaneously). Abbreviations: TBD: Transport Biofuels Directive 2003/30/EC of 08 May 2003; RED-I: Renewable Energy Directive 2009/28/EC from 23 April 2009 (in force); FQD: Fuel Quality Directive 2009/30/EC of 23 April 2009, amended 2016 (in force); ILUC: Directive (EU) 2015/1513 of 9 September 2015 (in force); RED-II: Renewable Energy Directive (recast) 2018/2001 of 11 December 2018 (in force).

TBD (2003)	RED-I (2009)	FQD (2009)	ILUC (2015)	RED-II (2018)
Replacing at least 5.75% of all transport fossil fuels (petrol and diesel) with biofuels by 2010. Intermediate target of 2% by 31 December 2005. Targets represent not mandatory reference values and refer to fuels for transport purposes placed on the market.	Renewable energy share in the transport sector at least 10% by 2020. Different renewable energy sources are factored in differently: the contribution of advanced biofuels counts 2 times and electricity from renewable sources 2.5 times.	Reduction in fuel GHG emissions of at least 6% by 2020.	Amends RED-I: Conventional biofuels smaller than 7% of final consumption of energy in transport in 2020 (cap). Advanced biofuels > 0.5% (voluntary). Harmonization of the list of feedstocks for biofuels eligible for double counting towards the 2020 target of 10% for renewable energy in transport (Annex IX).	Renewables: >14% of final energy consumption in transport in 2030. Advanced biofuels and biogas produced from feedstock listed in Part A of Annex IX: >0.2% of final energy consumption in transport in 2022; >1% in 2025; >3.5% by 2030. Biofuels and biogas from UCO and animal fat (Part B of Annex IX): <1.7% of transport fuels (cap). High ILUC biofuels: 0% in 2030. Definition of sustainability and GHG emission criteria for bioliquids used in transport to determine their eligibility for counting towards the overall 14% target and for financial support by public authorities.

proposal for the recast of the RED [77], launched shortly after, even if recognizing the limited future role of food crop-derived biofuels, "does not counter this assessment but stresses planning security for investors as the main argument for maintaining a transport sector target towards which first-generation biofuels can [still] contribute to a significant extent" [4] (p. 537). In other words, although the official justification for substituting oil-based fuels with biofuels is discredited, another one that of guaranteeing investor security—is used to cling to the original plan. Obviously, given that thus far the production of cellulose-based biofuels did not develop to any significant extent due to technological and economic hurdles, the first generation of biofuels, even if incompatible with the original justification narratives, was essential in keeping the policy discourse alive.

In the 2018 recast of the Renewable Energy Directive [24], the minimum share of renewable energy in the transport sector is increased to 14% by 2030; at least 3.5% of the transport energy must be derived from advanced biofuels and a progressive cap is placed on high ILUC biofuels until reaching 0% by 2030 (Table 1). Given the persistent difficulties in the development of advanced biofuels, in order to promote their development, the RED-II Directive retains the double-accounting system already proposed in RED-I. This means that the contribution of biofuels made from biomass not competing with food and not generating detrimental land-use changes (i.e., advanced biofuels listed in Part A of annex IX and other biofuels including used cooking oil in Part B of the annex) is double-counted in relation to the calculation of the mandatory share of renewable energy in the transport sector. Nonetheless, the production of biofuels from used cooking oil and animal fats (part B of Annex IX) is discouraged to less than 1.7% of the energy content of transport fuels supplied for consumption or use on the market. Hence, the recast of the RED hangs on to the original expectations about the supply of advanced cellulosic biofuels, and continues to promote their development.

A delegated act adopted by the Commission in May 2019 [78], supplementing the RED II, sets out the criteria for certification of low ILUC-risk biofuels, bioliquids and biomass fuels and for determining the high ILUC-risk feedstock for which a significant expansion of the production area into land with high-carbon stock is observed. With this act, palm oil diesel is considered 'high ILUC risk'.

Over time, we thus find not only a continuous increase in ambitions, but also a continuous complexification of the relation between the definition of justification narratives (the concerns to be addressed, i.e., the overall long-term goals) and the related normative narratives (the strategies and tactics adopted, i.e., the directives) resulting in a continuous patching of targets (short-term objectives).

4. Checking the plausibility of the narratives: Quantitative storytelling

Quantitative Storytelling is a novel approach that involves a quantitative exploration of multiple narratives in a given policy domain [79]. In the landscape of methodologies using stories, narratives or storytelling sketched by Moezzi et al. [80], quantitative storytelling positions itself as an approach that uses 'stories' as data. The stories used are government narratives on societal transitions. They are critically examined using predominantly quantitative methods, varying from integrated assessment models to simple back-of-the-envelope calculations. Rather than trying to compile evidence in support of a given narrative, or determine the 'best course of action', quantitative storytelling operates 'via negativa'; it attempts to test whether the examined framings are in conflict with quantitative analytical checks [81]. The theoretical foundation of quantitative storytelling is the theory of post-normal science, which explicitly acknowledges the unavoidable presence of scientific uncertainty and value plurality in the sustainability discussion [82,83]. Post-normal science was developed in response to 'wicked problems' [84] originating from societal concerns or policy issues for which a variety of legitimate and often contrasting perceptions about the 'truth' of knowledge claims co-exist, thus creating irreducible uncertainty, even ignorance, about the future state of the system. Postnormal science shifts the focus of quality control from the scientific information itself to the decision-making process and advocates the involvement of an extended peer community to check the quality of the narratives used to frame the problem.

Quantitative storytelling has first been systematically applied in the EU project 'Moving towards adaptive governance in complexity: Informing nexus security' (MAGIC) to check the robustness, the use-fulness and the fairness of the narratives used to discuss and select EU policies related to sustainability and climate change [35,81,85]. It has been applied to agriculture [86], desalination [87], aquaculture [88], and renewable energy sources (wind and solar) [89]. This is the first time quantitative storytelling is applied to biofuels.

To check the plausibility and the expected impacts of biofuels policy solutions, the conceptual map shown in Fig. 1 is useful. The rectangles in Fig. 1 describe the relations over functional elements, the ovals describe the expected technical coefficients (profiles of inputs and outputs) associated with the identity of the structural elements. This multi-level representation of the set of expected (established) relations among functional and structural elements allows a simultaneous handling of quantitative assessments across different scales and dimensions of analysis [88]. In the quantitative storytelling of biofuels presented in the next section, relevant aspects of the pattern of production and consumption of biofuels are assessed at different levels of analysis. Indeed, it is evident from the map that different aspects of the issue can only be 'seen' from different levels of analysis:

- (i) the whole network the overall requirement of liquid biofuels (expressed as a mix of biodiesel and bioethanol) is assessed on the basis of the end-uses in society to check whether the supply of biofuels is or can be consistent with a given justification narrative;
- (ii) any given functional elements of the network the overall characteristics and limits of the supply of a given system of biofuels, in order to check whether a given policy aimed at boosting the production of that system is credible (the plausibility of targets);
- (iii) any given components of a functional element the specific characteristics of relevant production processes are assessed to describe the "state of the art" technology for that product;
- (iv) any given structural elements of a component the individual profiles of inputs and outputs of specific local production processes are assessed to estimate local environmental impact.

The conceptual map can be used in a 'diagnostic mode' (examining the current situation) as well as in an 'anticipatory mode'. In the anticipatory mode, alternative options are explored by replacing selected elements with the desired alternatives.

Based on the relational organization of the information illustrated in Fig. 1, the compatibility of the expected outcomes of the policy—the expected establishment of a new metabolic pattern of production and consumption of biofuels— is evaluated against the expected constraints (across levels) on both biofuel production and consumption. These constraints may be: (i) beyond human control (feasibility); (ii) under human control (viability); (iii) determined by existing institutions and normative values (desirability). A systemic quantitative application has been illustrated by [25]. In this work, the conceptual map is used to perform four interrelated quality checks in the quantitative storytelling – feasibility, viability, desirability and openness, as described in [90].

1. *Feasibility* is essential for the debate on biofuels because external limits define the maximum amount that can be produced: it determines the plausibility of the normative narrative. Depending on the type of biofuels, external limits have to be assessed in different ways. For first-generation biofuels the external limits are given by: (i)



Fig. 1. The functional/structural relations of a biofuel production-consumption system.

the availability of land, water, soil, biodiversity, solar radiation needed to produce agro-feedstock; and (ii) the sink capacity required to handle the environmental pressure associated with the use of fertilizers and pesticides. The supply of cellulose-based advanced biofuels is limited by the availability of primary sources in the form of woody biomass. If this biomass is provided by crops, then they share the same set of limits with biofuels. In addition, if the biomass is derived from crop residues the limit is also associated with the need for leaving a fraction on the field for soil protection. The supply of biofuels produced from waste depends on the size of the waste flows available for their production. 2. The *viability* of specific processes of biofuel production depends on the availability of know-how and technology, the economic viability of the activity, which is affected by costs (labor requirement, fixed capital and inputs) and revenues, and the compliance with existent regulations. It is important to identify situations of *apparent* viability of biofuel production that depend on the existence of EU and/or national subsidies. The use of subsidies may be justified: (i) as a temporary solution to start a new viable/feasible process; (ii) as a stable cost for society that has to be justified by the expression of a required function.

J.J. Cadillo-Benalcazar et al.

- 3. Desirability refers to the consequences that the changes induced in the metabolic pattern of society (through the production and use of biofuels) have in terms of the concerns and quality of life of citizens (in relation to normative values). This includes potential indirect impacts on human health from pesticide and fertilizer use in feed-stock production, as well as the perceived benefits of the public investment of tax payers' money.
- 4. It is essential to consider the *degree of openness of the system* as imports can be used to lose-up existing constraints. The solution of externalizing requirements to other social-ecological systems carries potential consequences for the overall performance of the system. For example, externalization of the requirements of land, water, and labor and of environmental pressures may undermine the plausibility of the justifications narratives (policy objectives) concerned with energy security and reduction of emissions (e.g., because of indirect emissions due to land-use changes elsewhere). It also entails ethical concerns about the environmental and social impacts generated elsewhere, mainly in developing countries [9].

In the application of quantitative storytelling presented in this paper, only liquid biofuels are considered, as defined in Article 2 of the directive (EU) 2018/2001 [24] (p. 104). The data on the production, importation, exportation and consumption of biofuels in the EU and the Netherlands are from [91,92]. Data on the Dutch population was obtained from [93] and land use in agriculture in the Netherlands from [94]. Renewable energy data for the Dutch transport sector with simple and double counting were obtained from [95]. The Dutch Government [96] provides the composition and country of origin of the raw material used for the Production of biofuels that are delivered to the domestic market of the Netherlands. The reference year for the analysis of the Netherlands is 2017.

5. Results and discussion

5.1. Biofuel production and consumption in the Netherlands

The Netherlands was selected as a case study for the quantitative storytelling because it neatly demonstrates the discordance between justification, normative and explanation narratives. The Netherlands does not have a tradition in producing or using biofuels, like France or Germany, and has been relatively slow in developing biofuel market support mechanisms [97]. Nonetheless, its (physical) consumption of liquid biofuels in the transport sector steadily increased from 2 PJ in 2006 to 13 PJ in 2017 [95], and jumped to 21 PJ in 2018 [98] (p. 100). Whereas in 2017 the reported share of (all) renewables in transport amounted to only 5.9%, in 2018 it reached 9.6% (mainly due to the jump in biofuels use), thus bringing the 2020 target of 10% of the RED well within reach [99]. The majority of these reported renewables

consisted of liquid biofuels (75% in 2017 and 83% in 2018) [98] (p. 25). Severe land limitations (0.06 ha of arable land per capita) might have been expected to limit the contribution of biofuels to reaching the renewable energy target in transport. Nonetheless, in spite of the land constraint, the Netherlands not only significantly increased its use of biofuels in transport but is well on the way of becoming a major biofuel exporter. An apparent paradox?

The two main liquid biofuels used in the transport sector in the Netherlands are bioethanol, which is generally added to the regular gasoline, and biodiesel. The production, export and import of these two biofuels are illustrated for the period 2011–2018 in Fig. 2.

Bioethanol—The Netherlands does not produce significant quantities of bioethanol and imports this fuel as such. In 2017, the Netherlands imported around 6 PJ (Fig. 2), ranking third in Europe with the highest import volume after Germany (17 PJ) and the United Kingdom (11 PJ). On a per capita basis, the Netherlands occupied the first place with 0.4 GJ/per capita imported per year, followed by Germany and the United Kingdom (0.2 GJ/per capita per year). In 2017, all of the imported bioethanol concerned biofuels; most of the feedstock used for this imported bioethanol was produced within the EU. In particular, 46% of the bioethanol was produced from wheat, of which 87% from the EU; 33% from maize, of which 60% from the EU; 9% from sugar beet, of which 96% from the EU; 11% from sugar cane, entirely from Latin America; 1% from rye and triticale, nearly entirely from the EU [96].

Biodiesel—During the period 2008–2017, biodiesel production in the Netherlands increased by 23 times. In 2017, despite the low level of internal biodiesel consumption, the Netherlands was the country with the third-largest production of biodiesel in Europe (71 PJ), surpassed only by Germany (119 PJ) and France (92 PJ), countries with a considerably larger size. Since 2012 the Netherlands has surpassed the production of Spain and Italy, in spite of the smaller size of its economy and available cropland. According to Sheng Goh et al. [100], this production has been based mainly on imported feed-stocks (predominantly palm-oil). Since 2010 the production of biodiesel has been mainly destined for export. For example, in 2017, 59 PJ of biodiesel were exported, which represents 83% of the total production [91].

The destination of the biodiesel exported by the Netherlands during 2012–2018 is shown in Fig. 3. The data show a steady decrease in the volume of exports to Western Europe. This is mainly due to Germany reducing its imports after the year 2014 because of an increase in its internal supply. In contrast, Northern European countries increased the volume imported from the Netherlands, especially Sweden (the country with the third highest consumption of biofuels in road transportation in the EU in 2017 after France and Germany [91]) and Norway. According to Eurostat [92], Sweden increased its import of biodiesel from the Netherlands from 2014 to 2017 by 9 times. While the available data allow to monitor the type of biomass (feedstock) used in the production of biofuels that are consumed within the Netherlands and counting



Fig. 2. Supply (local production and import) and export of biodiesel and bioethanol in the Netherlands for the period 2011-2017 [91].



Fig. 3. Destination of biodiesel exports from the Netherlands in 2012–2018. Source: [92]

toward the renewable energy target in transport, this is not the case for the biofuels exported.

As noted earlier, recent implementation of the RED has encouraged a growing trend in the use of biofuels that allow double counting. According to [95], the consumption of biofuels eligible for double-counting increased from 3 PJ in 2009 to 7 PJ in 2017 in the Netherlands (real consumption without double counting). The consumption of conventionally produced biofuels, not eligible for double counting, decreased from 12 PJ in 2009 to 5 PJ in the same period. Hence, in 2017, 57% of the consumption concerned double-counted biofuels (entirely consisting of biodiesel) and 43% single-counted biofuels (entirely consisting of ethanol blended fuels). Notwithstanding the benefit of double-counting, the share of renewable energy in transport reported for 2017 (5.91% of which 4.55% corresponds to total biofuels and 1.36% to renewable electricity) fell short of the Dutch national target for that year (7.75%) [95,101].

The effect of the double counting on the reported biofuel use in the

transport sector of the Netherlands is shown in Fig. 4. The difference between double counting and net accounting corresponds to a virtual biofuel energy deployment that is mainly supplied by fossil fuels.

Apart from the questionable choice of solving real problems (how to reduce actual GHG emissions) with creative accounting solutions (by introducing double-counted imaginary savings subtracted from actual emissions), there is another question. How do we know if all the biodiesel that is double-counted in the bookkeeping effectively belongs to the category of biofuels that is eligible for double counting (those included in Part A and B of Annex IX of the RED) and does not include first-generation fuels?

According to [96], in 2017, biodiesel was the most commonly used biofuel in the Dutch transport sector (accounting for 57%). When considering the special category of biofuels eligible for inclusion in the double-counting scheme, the main raw material reported in the statistics is used cooking oil (UCO), representing around 90% of the total, and animal fats, representing around 10% of the total [96]. So it may be



Fig. 4. The effect of double counting on biofuel energy use reported in the transport sector in the Netherlands. Data . Source: [95]

concluded that the combination of these two raw materials – both eligible for double counting – constitutes the feed-stock used in the Netherlands to generate the (physical) (non-virtual) supply of the biofuels shown in Fig. 2. But is this possible?

Using official data from the Government of the Netherlands [96], in 2017, approximately 13 PJ (becoming 26 PJ with double accounting) of biodiesel from sources eligible for double-counting were delivered to the internal Dutch market. However, Eurostat [91] records that only 7 PJ (without double counting) were consumed in the transport system in that year. Possible explanations for this difference are the storage of this extra production in stocks or unrecorded exports. But could the Netherlands produce this quantity of double-countable biofuels? Based on the data, 90% of the actual flow of biodiesel (13 PJ) per year came from UCO [96]. This would require an availability of UCO in the Netherlands capable of producing 11.7 PJ per year, which corresponds to approximately 24 kg of UCO/person/year in terms of biomass. However, the UCO per capita collected in 2016 from household consumption was 0.21 kg/person/year [102]. Scaling up this value to the national level, this translates in 3600 tons of UCO collected on a year basis (ibid.). Considering the losses of conversion (about 25%, see [25]) and the energy content of biofuels (37.8 MJ/kg) we find that with the UCO gathered from the households in the Netherlands, one could only cover 100 TJ per year, less than one hundredth of the claimed production of biodiesel eligible for double counting. Note that UCO may also be collected from restaurants, collective kitchens and food industry, so the potential supply is larger. However, even considering these additional sources, the potential domestic supply could not reach >7 kg per person per year (assuming an UCO density of 0.91 kg/l) [102,103], considerably short of the hundreds of kg of diesel fuels consumed per capita per year. Hence, scaling up the biodiesel production from local food waste is simply not feasible; it is incompatible with local boundary conditions. Hence, it is the openness of the system that permits the current biofuel supply in the Netherlands: imported feedstock allows to overcome the local biophysical constraints.

As discussed earlier, the plausibility of the production of biofuels can be checked not only in terms of feasibility and openness but also in terms of viability. In 2017, the international price of biodiesel and bioethanol was higher than that of diesel and gasoline, by 1.8 and 1.4 times respectively [104]. In spite of the regulations and subsidies, several biofuel producing companies in the EU, including in the Netherlands, had to close within a few years of operation [105]. Hence, if biofuels were not subsidized and helped by regulation, they would not be economically viable. Clearly, the cost for society of these subsidies could be justified if they would help resolve the original concerns (e.g., GHG emissions, dependence on imports). However, if the expected enlargement of scale of biofuel production cannot possibly materialize to supply a significant amount of alternative fuels, it is unclear why the policy was selected in the first place. For a policy to be desirable it must be able to achieve the results strived for by the justification narratives (responsible use of taxpayers' money). The impromptu strategy of double accounting adopted by the Commission may temporarily hide the failure of previous normative narratives, but it does not solve the problem: if the proposed solution is implausible, cooking the books will not change the situation. On the contrary, it creates negative effects by distorting the biofuels market and in this way encourages frauds [106,107]. In the case of the Netherlands, the use of UCO for the production of biodiesel has increased significantly from 2011 to 2018 (see Fig. 5) and so has the number of different countries supplying it [108]. For example, while in 2015 the imported UCO originated from around 50 different countries, in 2017 this number had risen to 70 countries. This complicates controls on the authenticity of the biofuel feedstock reported and traded [106]. Indeed, there are suspicions that during the period 2015-2016, 59% of the biodiesel sold by the company that accounted for almost one third of the total Dutch biodiesel production in 2015, was erroneously certified as sustainable [109,110].

This simple exercise of quantitative story-telling shows an

incongruity between the official narratives about biofuels at EU level and the actual flows of biofuels observed at member state level in the Netherlands:

- In terms of feasibility, given the limited availability of land in the Netherlands, it is simply impossible to cover a significant share of liquid fuels for transportation from first-generation biofuels or cellulosic-based biofuels produced from *local* feedstock [111]. There are not sufficient local primary sources. The same is true for the production of biodiesel from UCO and animal fats. There is not sufficient *local* food waste.
- In terms of openness, the current deployment of biofuels is possible only because biofuels (bioethanol) and feed-stock (for biodiesel) are imported from elsewhere: not even 10% of the biofuels used in transport in the Netherlands is produced from local feedstock [108]. This questions the justification narratives of energy security (reducing the dependence on energy imports) and rural development. The observed increase in diversification of UCO suppliers in the Netherlands, while reducing dependence on any one supplier in particular, complicates the certification of the authenticity of feedstock supply and has encouraged fraud.
- In terms of viability, all current types of biofuels need financial support to survive on the fuels market, with required support being most pronounced for advanced biofuels [112]. As noted, most of the biodiesel produced in the Netherlands ends up being exported. Given that Rotterdam is a main entry point of international trade for the EU and biofuels and feedstock are bulk commodities, it makes eminent sense to do much processing/blending at ports. It is obvious that if support schemes guarantee a profit, feed-stocks will be imported and the biofuel produced exported, regardless of whether or not this responds to the original justification narratives.
- In terms of desirability, the fact that the production of conventional biofuels and biodiesel from UCO do not significantly contribute to the reduction of GHG emission, energy security or rural development in the Netherlands raises questions as to why the Dutch taxpayer should contribute to biofuel support schemes, especially if prone to fraud.

The case study of the Netherlands also shows that when implementing EU policies involving complex issues through national bureaucracies, the use of targets and simple indicators carries the risk of generating *displacement* [113]. National policy makers end up observing and judging results in relation to the 'artificial' representation of the problem given by the achievement of targets. They no longer appreciate the results in the context of the original justification narrative; the target becomes a justification in itself. The use of double-counting of UCOs and animal fats as well as statistical transfers of renewable energy-both facilitated by the RED (Articles 27 and 8, respectively)-to hit the renewable energy target in transport in the Netherlands [114-116] is a blatant example. The continuous patching of targets of the EU biofuel and renewable energy directives also distracts from the underlying concern (replacing liquid fossil fuels with biofuels to reduce GHG emissions and reliance on energy imports) and obfuscates the evaluation of the plausibility of the proposed policies. It is one issue whether or not it is possible to reach a temporary target (e.g., 3.5% of transport fuels from biofuels), but quite another whether or not this has led to a significant reduction in emissions and more energy security. The problem of displacement is not limited to biofuels policy but is evident in many policy areas in the EU (e.g., [117-120]) and elsewhere ([121]).

5.2. Biofuels at the EU level

In this section we provide the big picture at EU level based on the conceptual map shown in Fig. 1 and the Dutch case analyzed in section 5.1. Several key points emerge that question the formal justifications of EU biofuels policy.



Fig. 5. Relative increase in and origin of used cooking oil used for the production of biofuels in the Netherlands for the period 2011–2018 (based on physical energy content). . Source: [108]

The actual deployment of liquid biofuels (biodiesel and bioethanol) in the transport sector of the EU, compared to the use of liquid fossil fuels, is shown in Fig. 6. According to these data, the actual supply of biofuels in the EU would have to be increased by orders of magnitude in order to make a significant contribution toward the main goals stated in the justification narratives (energy security; reduction of emissions). But how can a significant increase in the actual supply of liquid biofuels possibly be realized when: (i) the RED II (normative narrative) limits the use of conventional biofuels to phase them out; (ii) advanced biofuels have not delivered on their promise; and (iii) 'other biofuels' based on food waste are not only limited in supply but also actively discouraged in the RED II? These data challenge the validity of the 'scientific evidence' used to defend the choice of current EU energy policy and flag a discrepancy between explanation and normative narratives.

The vast majority of the current (inadequate) supply of liquid biofuels (biodiesel and bioethanol) still consists in conventional biofuels (Fig. 7). This in spite of the fact that they have long been proven incompatible with the official justification narratives. Yet in 2016, biofuels still accounted for the vast majority of the subsidies paid in support of EU biofuel policy. As mentioned earlier, the slow adjustments in EU renewable energy policies could be defended as a way of protecting 'investor security' [4]. Nonetheless, in the long run, this justification may become politically unacceptable. Indeed, the continued production of conventional biofuel feedstock on cropland within the EU and the significant role played by biofuel feedstock (crops) imports (about 30–35% of feedstock for biodiesel and bioethanol is not of EU origin, see Fig. 7) question the desirability of the existing solution in face of the growing concern for future food and environmental security at both EU and global level [122–125].

For example, in 2016, the mix of biofuels in the transport sector of the EU (including UK) consisted of 80% biodiesel and 20% bioethanol [91]. Of the energy consumed as biodiesel, approximately 20% was derived from imported palm oil (around 91 PJ) (see Fig. 7) [126]. If the EU would replace the amount of biodiesel produced from imported palm oil (since 2019 palm oil diesel is considered 'high ILUC risk') with locally produced rapeseed, 1.8 million hectares of arable land would be required (based on an energy productivity of 52 GJ biodiesel/ha for rapeseed compared to 88 GJ biodiesel/ha for oil palm [127]). Note that these 1.8 million hectares cultivated with rapeseed would cover only 1% of the total diesel consumed in the EU, but significantly increase the tension between the Common Agricultural Policy objectives of increasing competitiveness and preserving landscapes and biodiversity [128].

As shown in Fig. 7, the contribution of cellulosic bioethanol is practically irrelevant and has failed to meet the original expectations raised by the Commission [45] (p. 37–39) to overcome the biophysical limits on the supply side. The supposition that any type of woody-biomass (including energy crops, wood and any other forms of biomass) can be used as feedstock for biofuel production resulted in a dramatic overestimation of the potential supply from cellulosic biofuels



Fig. 6. Energy used in the EU transport sector (EU27 plus UK) in 2009–2017 by source: fossil fuels (diesel and gasoline) versus biofuels (biodiesel and bioethanol) [91].



Fig. 7. Feedstock used for the production of the biofuels used in the European Union, according to its origin (external versus locally produced), in the year 2016 (data).

Source: [126]

and has played an important role in keeping biofuels alive in the policy discussion. The production of cellulosic ethanol has not fulfilled expectations because of technical and economic problems (viability) [111,129] and falls short even of minimum targets [5].

As for the production of biodiesel, only 29% is produced from feedstock listed in Annex IX eligible for double counting. However, these biofuels are based on the utilization of resources of limited availability: (i) 8% from animal fat (a by-product the production of which cannot be scaled up easily and the use of which is discouraged in RED II) and 3% from tall oil (a co-product of the production of wood pulp in northern European countries); and (ii) 18% from UCO, the limitations and the problems of which have been pointed out for the case study of the Netherlands. Biodiesel from local UCO could at most replace 1.5% of EU diesel consumption (2015 level) [130]. Note that the use of biomass waste as feedstock for biofuel generation carries the risk of discouraging the production of 'higher-value' materials [131]. A central pillar of the EU bioeconomy strategy is the principle of 'cascading use of biomass' that prioritizes exploitation towards added value products before eventual recycling and conversion to energy [132]. For instance, in order to meet RED II targets, it is expected that an increasing fraction of tall oil production will be diverted for the generation of advanced biofuels [133]. This could potentially limit its use for the production of biobased chemicals and materials.

The current reliance on used cooking oil for the supply of liquid fuels is indicative of the absurdity of the situation. Given that UCO is only available in small quantities (about 7 kg per person per year–[103]) and difficult and expensive to collect, import of UCO cannot be considered a "sustainable solution" to guarantee energy security and reduce emissions (e.g., [116]). Indeed, even the Commission seems to recognize this: Article 27 of RED II states that "for the calculation of the numerator, the share of biofuels and biogas produced from the feedstock listed in part B of Annex IX shall, except for in Cyprus and Malta, be limited to 1.7% of the energy content of transport fuels supplied for consumption or use on the market". Thus, remarkably, UCO is still eligible for double counting to promote its use but at the same time its use is explicitly discouraged.

This shows again that the plausibility of biofuels policies boils down to the credibility of a large-scale cellulose-based fuel revolution. The technology necessary to achieve commercial development of advanced biofuels has yet to overcome several obstacles [134], and the doubts about the plausibility of the proposed solutions are growing [135].

A holistic picture of the pattern of production and end-uses of

biofuels by category provides a better understanding of the different issues involved in biofuels policy. Indeed, we must keep in mind that targets in themselves may have nothing to do with the expectations of the underlying justification narrative. Modest intermediate targets may represent a good start for a transition to a low carbon economy, but only if the proposed solution is good and can grow to significant levels. If we cannot expect advanced biofuels to cover a significant fraction of the total fuel supply in the medium-long term, then we should not be very excited by the reaching of an initial target.

In Section 3 we saw that three 'official' justification narratives have been put forward in the EU discourse on biofuels policy: (i) solving the problem of energy security; (ii) solving the problem of greenhouse gas emissions in the transportation sector; (iii) stimulating rural development. Other justifications are implicit, such as solving temporary problems in the implementation of agricultural policies (e.g., creating market outlets) and guaranteeing investor security [3,4]. By falsifying official narratives, quantitative storytelling seeks to broaden the discussion and draw attention to other ('untold') perceptions of a given issue that may exist among other social actors. For example, with regard to biofuels, the following justifications supported by different types of social actors may have played and still play a role in the choice of biofuels policies:

- 1. Business sector: (i) Farmers may see another outlet for products and a justification for continued subsidies; (ii) Agribusinesses may see market expansion for agricultural inputs and diversification (guaranteed markets) for outputs, as well as a narrative supporting industrialized agriculture (increasingly challenged by consumers, but the only model capable of supplying biofuel feedstock at scale); (iii) The car and oil industries may play along as biofuels greenwash their business model and existing technology (Internal combustion engine running on liquid fuels), which are being increasingly challenged by concerns about climate change and resource exhaustion. They may prefer the trouble of integrating a token amount of biofuel to the alternative of higher fuel economy standards, higher car/road/ fuel taxes, increased pressure towards other modes of transport, or other drivetrain technologies, such as electric cars.
- 2. EU consumers: Consumers' environmental conscience is salved if they can believe they have done "something" about transport emissions. The use of biofuels permits them to do so without having to restrict the use of their cars.

3. From EU politicians: they are forced to adopt discourses and narratives that please their voters.

All these perceptions and concerns are not necessarily equivalent in terms of the robustness of the underlying knowledge claims, but, at the level of the perceptions of social actors, equally legitimate. When dealing with complex problems, it is unavoidable that the adoption of a hegemonic narrative focusing only on a limited number of 'official concerns' (criteria of performance) and a closed set of options (chosen alternatives) generates hypocognition [136,137], i.e., the potentially dangerous (and possibly intentional) neglect of other potentially relevant concerns and useful alternatives in the definition of the option space. The ultimate goal of quantitative storytelling is to expose other justification narratives that may be in play and to dismantle 'socially constructed ignorance' [113].

6. Conclusions and policy implications

The quantitative storytelling presented points at the inadequacy of the explanation narratives currently used to inform EU decision makers in the policy domain of biofuels. The explanation narratives lack the systemic thinking that would be required for structuring the quantitative analysis of the metabolic pattern of biofuel production and consumption in order to suggest effective and meaningful targets. The embarrassing assumption that the first generation of agro-biofuels represents a feasible, viable and desirable alternative for replacing a significant fraction of liquid fossil fuels while reducing emissions simply indicates a poor semantic framing of the quantitative analysis.

In the case of the Netherlands, biofuels are produced from imported feedstock and then exported. This behavior is inconsistent with the main justification narratives and could have been anticipated given the implausibility of the policy: in the Netherlands there is not enough land nor sufficient wastes (UCO included) to produce the required amount of feed-stock. In this situation, the justification narratives of energy security and reduction of emissions (considering the effect of land-use change of imported inputs) definitely do not hold.

Indeed, our analysis shows that: (i) The official set of justification narratives does not refer to concerns that are addressed by the selected policies. The existing policies seem to address other legitimate concerns of specific groups of relevant social actors (citizens, consumers, entrepreneurs, politicians). Quantitative story-telling is useful to falsify official storytelling and points at the potential role of other (hidden) justification narratives that may be less defendable in public discussions; (ii) The targets proposed by the various overlapping directives are implausible. Not only have they been proven implausible by a simple quantitative exercise, but also by the systemic missing of the targets in time [4,5].

The existing confusion in the policy discourse has allowed a dangerous twist in the logic of target setting. The expected direction of the chain of policy decisions is: (i) why do we need biofuels? (the concerns to be addressed-justification narrative); (ii) how can we achieve these goals? ('scientific evidence'-explanation narrative) (iii) what actions should we undertake toward achieving these goals? (directives/ targets-normative narrative). In a situation of extreme confusion, as is the case with biofuels policy, there is the risk of inverting this order and cutting corners. Depending on their specific legitimate perspectives, social actors may first identify a convenient target to set (or preserve) and then select a fitting justification (from among the many possible ones) to support that target. This phenomenon appears to be common in the policy domain of biofuels. Indeed, Skogstad and Wilder [5] (p. 350) claim that ambiguity is essential to create "opportunities for institutional agenda-setters to build coalitions in support of biofuel mandates/ targets". A responsible process of policy making should not be based on the coupling of targets with justification narratives. Instead, the focus should be on the congruence between justification narratives (what are the concerns to be addressed) and normative narratives (the identification of what should be done) based on plausible explanations. A check on the plausibility of the proposed solution is essential. Setting policy targets on the basis of implausible solutions will not help solve the original concern and only generate (additional) confusion.

While this paper shows that biofuels remain a wicked problem [84] for decision makers, it cannot be denied that the various policy strategies and responses of member states did create a profitable biofuel industry as well as an arcane object of study for scientists¹ (thus adding a new category of social actors having a stake in the discussion).

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

This paper has greatly benefited from the detailed comments and constructive suggestions of two anonymous reviewers. The authors gratefully acknowledge funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No. 689669 (MAGIC) and the government of Catalonia (AGAUR) under grant agreement 2017-SGR-230 (IASTE). The Institute of Environmental Science and Technology (ICTA) has received financial support from the Spanish Ministry of Science, Innovation and Universities (MICINN) through the "María de Maeztu" program for Units of Excellence (CEX2019-000940-M). This work reflects only the authors' view; the funding agencies are not responsible for any use that may be made of the information it contains.

References

- M. Banja, R. Sikkema, M. Jégard, V. Motola, J.F. Dallemand, Biomass for energy in the EU – the support framework, Energy Policy 131 (2019) 215–228, https:// doi.org/10.1016/j.enpol.2019.04.038.
- [2] K. Janda, L. Kristoufek, D. Zilberman, Biofuels: Policies and impacts, Agric. Econ. (Czech Republic) 58 (2012) 372–386. https://doi.org/10.17221/124/2011-agr icecon.
- [3] M. Londo, E. Deurwaarder, Developments in eu biofuels policy related to sustainability issues: overview and outlook, Biofuels, Bioprod. Bioref. 1 (2007) 292–302, https://doi.org/10.1002/bbb.40.
- [4] A. Purkus, E. Gawel, D. Thrän, The role of a renewable energy target for the transport sector beyond 2020: Lessons learned from eu biofuel policy, in: E. Gawel, S. Strunz, P. Lehmann, A. Purkus (Eds.), Eur. Dimens. Ger. Energy Transit. Oppor. Conflicts, Springer International Publishing, 2019: pp. 527–542. https ://doi.org/10.1007/978-3-030-03374-3_30.
- [5] G. Skogstad, M. Wilder, Strangers at the gate: the role of multidimensional ideas, policy anomalies and institutional gatekeepers in biofuel policy developments in the USA and European Union, Policy Sci. 52 (2019) 343–366, https://doi.org/ 10.1007/s11077-019-09351-5.
- [6] S. Stattman, A. Gupta, L. Partzsch, P. Oosterveer, Toward Sustainable Biofuels in the European Union? Lessons from a decade of hybrid biofuel governance, Sustainability 10 (2018) 4111, https://doi.org/10.3390/su10114111.
- [7] L. Di Lucia, Too difficult to govern? An assessment of the governability of transport biofuels in the EU, Energy Policy 63 (2013) 81–88, https://doi.org/ 10.1016/j.enpol.2013.08.080.
- [8] G. de L.T. Oliveira, B. McKay, C. Plank, How biofuel policies backfire: misguided goals, inefficient mechanisms, and political-ecological blind spots, Energy Policy 108 (2017) 765–775. https://doi.org/10.1016/j.enpol.2017.03.036.
- [9] C. Hinkes, Adding (bio)fuel to the fire: discourses on palm oil sustainability in the context of European policy development, Environ. Dev. Sustain. (2019) 1–22, https://doi.org/10.1007/s10668-019-00541-y.
- [10] B. Kovarik, Henry Ford, Charles Kettering and the fuel of the future, Automot. Hist. Rev. Spring (1998) 7–27.
- [11] B. Kovarik, Biofuels in history., in: B.P. Singh (Ed.), Biofuel Crop. Prod. Physiol. Genet., CABI, Wallingford, 2013: pp. 1–22. https://doi.org/10.1079/978184593 8857.0001.
- [12] S. Brody, Bioenergetic And Growth, Hafner Publishing Company, New York, 1945.

 $^{^1}$ In the period 2000–2018, the number of publications about biofuel increased from 100 to >5650 (using 'biofuel' as a search term in Scopus, a database containing 18,000 published journals).

- [13] D. Pimentel, M.A. Moran, S. Fast, G. Weber, R. Bukantis, L. Balliett, P. Boveng, C. Cleveland, S. Hindman, M. Young, Biomass energy from crop and forest residues, Science 212 (1981) 1110–1115, https://doi.org/10.1126/ science.212.4499.1110.
- [14] W. Lockeretz, Crop residues for energy: comparative costs and benefits for the farmer, the energy facility, and the public, Energy Agric. 1 (1981) 71–89, https:// doi.org/10.1016/0167-5826(81)90007-3.
- [15] V. Smil, Biomass Energies, Springer US, Boston, MA, 1983. https://doi.org/ 10.1007/978-1-4613-3691-4_1.
- [16] D.J.C. MacKay, Sustainable Energy Without the Hot Air, UIT Cambridge Ltd., Cambridge, UK, 2009. http://www.inference.org.uk/sustainable/book/tex/sew tha.pdf.
- [17] Z. Kovacic, R. Strand, T. Völker, The Circular Economy in Europe: Critical Perspectives on Policies and Imaginaries, Routledge, 2020.
- [18] S. Jasanoff, S.-H. Kim, Sociotechnical imaginaries and national energy policies, Sci. Cult. (Lond) 22 (2013) 189–196, https://doi.org/10.1080/ 09505431.2013.786990.
- [19] S. Jasanoff, S.-H. Kim, Dreamscapes of Modernity Sociotechnical Imaginaries and the Fabrication of Power, University of Chicago Press, 2015. https://doi.org /10.7208/chicago/9780226276663.001.0001.
- [20] M. Giampietro, S.O. Funtowicz, From elite folk science to the policy legend of the circular economy, Environ. Sci. Policy. 109 (2020) 64–72, https://doi.org/ 10.1016/i.envsci.2020.04.012.
- [21] S. Funtowicz, J.R. Ravetz, Emergent complex systems, Futures 26 (1994) 568–582, https://doi.org/10.1016/0016-3287(94)90029-9.
- [22] K.B. Janda, M. Topouzi, Telling tales: using stories to remake energy policy, Build. Res. Inf. 43 (2015) 516–533, https://doi.org/10.1080/ 09613218.2015.1020217.
- [23] U. Felt, B. Wynne, Taking European Knowledge Society Seriously. Report of the Expert Group on Science and Governance to the Science, Economy and Society Directorate, Directorate-General for Research, European Commission. EUR 22700 – Science & Governance, Office for Official Publications of the European Communities, Luxembourg, 2007. https://ec.europa.eu/research/science-societ y/document library/pdf 06/european-knowledge-society en.pdf.
- [24] European Commission, Directive (EU) 2018/2001 of the European Parliament and of The Council of 11 December 2018 on the promotion of the use of energy from renewable sources (recast), Off. J. Eur. Union. L328 (2018) 82–209. https://eur-lex.europa.eu/eli/dir/2018/2001/oi.
- [25] M. Ripa, J.J. Cadillo-Benalcazar, M. Giampietro, Cutting through the biofuel confusion: an analytical framework to check the feasibility, viability and desirability of biofuels, Energy Strateg. Rev. Submitted (2020).
- [26] European Commission, Knowledge for Policy | Bioeconomy | Glossary, (2020). https://ec.europa.eu/knowledge4policy/bioeconomy/glossary_en (accessed June 19, 2020).
- [27] FAO, Definitions: Energy Bioenergy terminology. Webpage of the Food and Agriculture Organization of the United Nations, (2020). http://www.fao.org/ener gy/home/definitions/en/ (accessed June 19, 2020).
- [28] European Parliament and Council of the European Union, Directive (EU) 2015/ 1513 of the European Parliament and of the Council of 9 September 2015 amending Directive 98/70/EC relating to the quality of petrol and diesel fuels and amending Directive 2009/28/EC on the promotion of the use of energy from renewable sources, Off. J. Eur. Union. L 239 (2015) 1–29. https://eur-lex.europa. eu/legal-content/EN/TXT/?uri=CELEX:32015L1513.
- [29] European Commission, Indirect Land Use Change (ILUC). Memo, 17 October 2012 (last update 12-02-2015), (2012). https://ec.europa.eu/commission/press corner/detail/en/MEMO 12 787.
- [30] US Congress, Energy Independence and Security Act of 2007 (P.L. 110-140; EISA). H.R. 6 (110th)., (2007). https://www.govinfo.gov/content/pkg/ BILLS-110hr6enr/pdf/BILLS-110hr6enr.pdf.
- [31] UNICA, EPA reaffirms sugarcane biofuel is advanced Renewable fuel with 61% less emissions than gasoline., Newsl. Brazilian Sugarcane Ind. Assoc. 02 March 2010. (2010). http://english.unica.com.br/news/38990375920334398749/epareaffirms-sugarcane-biofuel-is-advanced-renewable-fuel-with-61-por-cento-less-e missions-than-gasoline/ (accessed June 19, 2020).
- [32] EPA, Regulation of Fuels and Fuel Additives: Changes to Renewable Fuel Standard Program; Final Rule. 40 CFR Part 80. Federal Register, Vol. 75, No. 58, Friday, March 26, 2010, Rules and Regulations., (2010). https://www.govinfo. gov/content/pkg/FR-2010-03-26/pdf/2010-3851.pdf.
- [33] A. Meijide, C. de la Rua, T. Guillaume, A. Röll, E. Hassler, C. Stiegler, A. Tjoa, T. June, M.D. Corre, E. Veldkamp, A. Knohl, Measured greenhouse gas budgets challenge emission savings from palm-oil biodiesel, Nat. Commun. 11 (2020) 1–11, https://doi.org/10.1038/s41467-020-14852-6.
- [34] A.M. Hennecke, M. Faist, J. Reinhardt, V. Junquera, J. Neeft, H. Fehrenbach, Biofuel greenhouse gas calculations under the European Renewable Energy Directive – a comparison of the BioGrace tool vs. the tool of the Roundtable on Sustainable Biofuels, Appl. Energy 102 (2013) 55–62, https://doi.org/10.1016/j. apenergy.2012.04.020.
- [35] A. Saltelli, M. Giampietro, The Fallacy of Evidence-Based Policy, in: A. Benessia, S. Funtowicz, M. Giampietro, Â. Guimarães Pereira, J.R. Ravetz, A. Saltelli, R. Strand, J.P. van der Sluijs (Eds.), Sci. Verge, Consortium for Science, Policy & Outcomes, Tempe, Ariz., 2016, pp. 31–69.
- [36] J.M. Amezaga, S.L. Boyes, J.A. Harrison, Biofuels policy in the European Union, in: 7th Int. Biofuels Conf. 11-12 Febr. 2010, New Delhi, Winrock International India, 2010. https://eprint.ncl.ac.uk/174992.

- [37] European Commission, Biofuels Strategy: Background Memo/06/65, Brussels, 8 February 2006, (2006). https://ec.europa.eu/commission/presscorner/detail/e n/MEMO 06 65.
- [38] P. Bocquillon, A. Evrard, Complying with, resisting or using Europe? Explaining the uneven and diffuse Europeanization of French renewable electricity and biofuels policies, in: I. Solorio, H. Jörgens (Eds.), A Guid. to EU Renew. Energy Policy, Edward Elgar Publishing, Cheltenham, UK, 2017: pp. 162–182. https:// doi.org/10.4337/9781783471560.00019.
- [39] V. Dossche, S. Ozinga, When the Solution is the Problem: The EU and its Policies on Agrofuels. FERN Briefing Note 01 (Bioenergy and forests), 08 July 2008, 2008. https://www.fern.org/news-resources/when-the-solution-is-the-problem-the-euand-its-policies-on-agrofuels-682/ (accessed June 21, 2020).
- [40] European Commission, Commission staff working document Annex to the Communication from the Commission to the Council and the European Parliament – Towards a sustainable European wine sector – Impact assessment. {COM(2006) 319} */ SEC/2006/0770 */, (2006). https://eur-lex.europa.eu/l egal-content/EN/TXT/HTML/?uri=CELEX:52006SC0770&from=ET.
- [41] European Commission, Commission Regulation (EEC) No 334/93 of 15 February 1993 laying down detailed implementing rules for the use of land set aside for the provision of materials for the manufacture within the Community of products not primarily intended for human or animal, (1993). https://eur-lex.europa.eu /legal-content/EN/TXT?/uri=CELEX%3A31993R0334.
- [42] The Council of the European Union, Council Regulation (EC) No 1782/2003 of 29 September 2003 establishing common rules for direct support schemes under the common agricultural policy and establishing certain support schemes for farmers and amending Regulations (EEC) No 2019/93,, (2003). https://eur-lex.eu ropa.eu/legal-content/EN/TX1/?uri=CELEX%3A32003R1782.
- [43] European Commission, COM(2001) 547. Comm. from the Commission to the European Parliament, the Council, the Economic and Social Committee and the Committee of the Regions on alternative fuels for road transportation and on a set of measures to promote the use of biofuels., (2001). https://ec.europa.eu/transpa rency/regdoc/rep/1/2001/EN/1-2001-547-EN-F1-1.Pdf.
- [44] European Commission, Green Paper Towards a European strategy for the security of energy supply. COM/2000/0769 final., Brussels, 2000. https://eur-le x.europa.eu/legal-content/EN/TXT/?uri=celex:52000DC0769.
- [45] European Commission, Energy for the future: renewable sources of energy. White paper for a ccommunity strategy and action plan. COM (97) 599 final., Brussels, 1997. https://europa.eu/documents/comm/white_papers/pdf/com97_599_en. pdf.
- [46] The Council of the European Union, Council Regulation (EC) No 1698/2005 of 20 September 2005 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD), Off. J. Eur. Union. L–277 (2005) 1–40. https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex%3A32013R1305.
- [47] European Parliament and Council of the European Union, Regulation (EU) No 1305/2013 of the European Parliament and of the Council of 17 December 2013 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD) and repealing Council Regulation (EC) No 1698/2005, Off. J. Eur. Union. L 347 (2013) 487–548. https://eur-lex.europa.eu/legal-conten t/EN/TXT/?uri=cclex%3A32013R1305.
- [48] European Parliament and Council of the European Union, Directive 2003/30/EC of the European Parliament and of the Council of 8 May 2003 on the promotion of the use of biofuels or other renewable fuels for transport. COD 2001/0265., (2003). https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=CELEX%3A3 2003L0030.
- [49] European Commission, Biofuels Progress Report. Report on the progress made in the use of biofuels and other renewable fuels in the Member States of the European Union. COM(2006) 845 final ., (2007). https://eur-lex.europa.eu/Le xUriServ/LexUriServ.do?uri=COM:2006:0845:FIN:EN:PDF.
- [50] European Commission, Biomass action plan {SEC(2005) 1573} /* COM/2005/ 0628 final */, (2005). https://eur-lex.europa.eu/legal-content/EN/TXT/? uri=celex:52005DC0628.
- [51] European Commission, An EU Strategy for Biofuels {SEC(2006) 142} /* COM/ 2006/0034 final */, (2006). https://eur-lex.europa.eu/legal-content/EN/TXT/? uri=celex:52006DC0034.
- [52] European Parliament and Council of the European Union, Directive 2009/28/EC of the European Parliament and of the Council of 23 April 2009 on the promotion of the use of energy from renewable sources and amending and subsequently repealing Directives 2001/77/EC and 2003/30/EC (Text with EEA relevance), (2009) 1–47.
- [53] European Commission, Renewable Energy Road Map Renewable energies in the 21st century: building a more sustainable future. COM/2006/0848 final, (2007). https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX:52006DC0848.
 [54] European Commission, Directive 2009/30/EC of the European Parliament and of
- [54] European Commission, Directive 2009/30/EC of the European Parliament and of the Council of 23 April 2009 amending Directive 98/70/EC as regards the specification of petrol, diesel and gas-oil and introducing a mechanism to monitor and reduce greenhouse gas emissions and amend, Off. J. Eur. Union. L140 (2009) 88–113. https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex%3A3 2009L0030 (accessed April 4, 2020).
- [55] P. Thompson, The agricultural ethics of biofuels: the food vs fuel debate, Agriculture 2 (2012) 339–358, https://doi.org/10.3390/agriculture2040339.
- [56] N. Alexandratos, J. Bruinsma, World Agriculture towards 2030/2050: the 2012 revision, Rome, 2012. www.fao.org/economic/esa (accessed June 28, 2020).
- [57] D.J. Tenenbaum, Food vs. Fuel, Diversion of crops could cause more hunger, Environ. Health Perspect. 116 (2008), https://doi.org/10.1289/ehp.116-a254.
- [58] M. Giampietro, K. Mayumi, The biofuel delusion: the fallacy of large scale agrobiofuels production, Earthscan-Routledge (2009).

- [59] A. Chakrabortty, Secret report: biofuel caused food crisis., Guard. 3 July 2008. (2008). https://www.theguardian.com/environment/2008/jul/03/biofuels.rene wableenergy (accessed June 28, 2020).
- [60] J. Fargione, J. Hill, D. Tilman, S. Polasky, P. Hawthorne, Land clearing and the biofuel carbon debt, Science (80-.). 319 (2008) 1235–1238. https://doi.org/1 0.1126/science.1152747.
- [61] T. Searchinger, R. Heimlich, R.A. Houghton, F. Dong, A. Elobeid, J. Fabiosa, S. Tokgoz, D. Hayes, T.-H. Yu, Use of U.S. Croplands for biofuels increases greenhouse gases through emissions from land-use change, Science (80-.). 319 (2008) 1238–1241. https://doi.org/10.1126/science.1151861.
- [62] D.F. Correa, H.L. Beyer, J.E. Fargione, J.D. Hill, H.P. Possingham, S.R. Thomas Hall, P.M. Schenk, Towards the implementation of sustainable biofuel production systems, Renew. Sustain. Energy Rev. 107 (2019) 250–263.
- [63] M. Giampietro, S. Ulgiati, Integrated assessment of large-scale biofuel production, CRC. Crit. Rev. Plant Sci. 24 (2005) 365–384, https://doi.org/10.1080/ 07352680500316300.
- [64] M. Giampietro, S. Ulgiati, D. Pimentel, Feasibility of large-scale biofuel production, Bioscience 47 (1997), https://doi.org/10.2307/1313165.
- [65] H.K. Gibbs, M. Johnston, J.A. Foley, T. Holloway, C. Monfreda, N. Ramankutty, D. Zaks, Carbon payback times for crop-based biofuel expansion in the tropics: the effects of changing yield and technology, Environ. Res. Lett. 3 (2008), https:// doi.org/10.1088/1748-9326/3/3/034001.
- [66] R. Edwards, D. Mulligan, L. Marelli, Indirect Land Use Change From Increased Biofuels Demand – Comparison of Models and Results for Marginal Biofuels Production from Different Feedstocks. EUR Scientific and Technical Research Report No. JRC59771., Publications Office of the European Union, 2010. http ://publications.jrc.ec.europa.eu/repository/handle/JRC59771.
- [67] R.J. Plevin, M. O'Hare, A.D. Jones, M.S. Torn, H.K. Gibbs, Greenhouse gas emissions from biofuels' indirect land use change are uncertain but may be much greater than previously estimated, Environ. Sci. Technol. 44 (2010) 8015–8021, https://doi.org/10.1021/es101946t.
- [68] Anonymous, UN independent rights expert calls for five-year freeze on biofuel production, UN News, 26 Oct. 2007. (2007). https://news.un.org/en/story /2007/10/237362-un-independent-rights-expert-calls-five-year-freeze-biofuel-p roduction (accessed February 12, 2019).
- [69] T. Gomiero, Large-scale biofuels production: a possible threat to soil conservation and environmental services, Appl. Soil Ecol. 123 (2018) 729–736, https://doi. org/10.1016/j.apsoil.2017.09.028.
- [70] European Commission, COM (2012) 595 final: Proposal for a Directive of the European Parliament and of the Council amending Directive 98/70/EC relating to the quality of petrol and diesel fuels and amending Directive 2009/28/EC on the promotion of the use of energy from renewa, (2012). https://www.europarl. europa.eu/registre/docs_autres_institutions/commission_europeenne/com/2012/ 0595/COM_COM(2012)0595_FR.pdf.
- [71] L. Asveld, The need for governance by experimentation: the case of biofuels, Sci. Eng. Ethics 22 (2016) 815–830, https://doi.org/10.1007/s11948-015-9729-y.
- [72] Euroactive, Biofuels industry sent 'three mails an hour' in ILUC lobby offensive EURACTIV.com (2013).
- [73] C. Charles, I. Gerasimchuk, R. Bridle, T. Moerenhout, E. Asmelash, T. Laan, Biofuels: At What Cost? A review of costs and benefits of EU biofuel policies, 2013. https://www.iisd.org/gsi/sites/default/files/biofuels_subsidies_eu_review. pdf.
- [74] C. Charles, I. Gerasimchuk, R. Bridle, T. Moerenhout, E. Asmelash, T. Laan, Addendum to Biofuels—At What Cost? A review of costs and benefits of EU biofuel policies, 2013. https://www.iisd.org/gsi/sites/default/files/biofuels_eu addendum.pdf.
- [75] European Commission, Proposal for a Directive of the European Parliament and of the Council, amending Directive 98/70/EC relating to the quality of petrol and diesel fuels and amending Directive 2009/28/EC on the promotion of the use of energy from renewable sources, 2012.
- [76] Transport & Environment, What's the problem and how can we fix it?, (2020). htt ps://www.transportenvironment.org/what-we-do/biofuels-whats-problem (accessed August 24, 2020).
- [77] Council of the European Union, Proposal for a directive of the European Parliament and of the Council on the promotion of the use of energy from renewable sources (recast) - Guidance for further work. 8697/2/17 REV 2, (2017). http://data.consilium.europa.eu/doc/document/ST-8697-2017-REV -2/en/pdf.
- [78] European Commission, Commission Delegated Regulation (EU) 2019/807 of 13 March 2019 supplementing Directive (EU) 2018/2001 of the European Parliament and of the Council as regards the determination of high indirect landuse change-risk feedstock for which a significant expans, Off. J. Eur. Union. L 133 (2019) 1–7. http://data.europa.eu/eli/reg_del/2019/807/oj.
- [79] Wikipedia, Quantitative storytelling, (n.d.). https://en.wikipedia.org/wiki /Quantitative_storytelling (accessed April 3, 2020).
- [80] M. Moezzi, K.B. Janda, S. Rotmann, Using stories, narratives, and storytelling in energy and climate change research, Energy Res. Soc. Sci. 31 (2017) 1–10, https://doi.org/10.1016/j.erss.2017.06.034.
- [81] A. Saltelli, M. Giampietro, What is wrong with evidence based policy, and how can it be improved? Futures 91 (2017) 62–71, https://doi.org/10.1016/j. futures.2016.11.012.
- [82] S. Funtowicz, J. Ravetz, M. O'Connor, Challenges in the use of science for sustainable development, Int. J. Sustain. Dev. 1 (1998) 99–107, https://doi.org/ 10.1504/IJSD.1998.004206.
- [83] S. Funtowicz, J. Ravetz, Uncertainty and Quality in Science for Policy, Kluwer Academic Publishers, Dordrecht, 1990.

- [84] H.W.J. Rittel, M.M. Webber, Dilemmas in a general theory of planning, Policy Sci. 4 (1973) 155–169, https://doi.org/10.1007/BF01405730.
- [85] K.B. Matthews, K.L. Blackstock, M. Rivington, K. Waylen, D.G. Miller, D. Wardell-Johnson, Z. Kovacic, A. Renner, M. Ripa, M. Giampietro, Delivering more than the "Sum of the Parts": using Quantitative Storytelling to address the challenges of conducting science for policy in the EU land, water and energy nexus, in: 22nd Int. Congr. Model. Simulation, Hobart, Tasmania, Aust. 3 to 8 December 2017, Australia, 2017, pp. 15–21.
- [86] M. Giampietro, Anticipation in Agriculture, in: R. Poli (Ed.), Handb. Anticip., Springer International Publishing, Cham, 2018, pp. 1–35. https://doi.org/10.1 007/978-3-319-31737-3_23-1.
- [87] T. Serrano-Tovar, B. Peñate Suárez, A. Musicki, J.A. de la Fuente Bencomo, V. Cabello, M. Giampietro, Structuring an integrated water-energy-food nexus assessment of a local wind energy desalination system for irrigation, Sci. Total Environ. 689 (2019) 945–957, https://doi.org/10.1016/j.scitotenv.2019.06.422
- [88] J.J. Cadillo-Benalcazar, M. Giampietro, S.G.F. Bukkens, R. Strand, Multi-scale integrated evaluation of the sustainability of large-scale use of alternative feeds in salmon aquaculture, J. Clean. Prod. 248 (2020), 119210, https://doi.org/ 10.1016/j.jclepro.2019.119210.
- [89] A. Renner, M. Giampietro, Socio-technical discourses of European electricity decarbonization: contesting narrative credibility and legitimacy with quantitative story-telling, Energy Res. Soc. Sci. 59 (2020), 101279, https://doi.org/10.1016/j. erss.2019.101279.
- [90] M. Giampietro, J.J. Cadillo Benalcazar, L.J. Di Felice, M. Manfroni, L. Pérez Sánchez, A. Renner, M. Ripa, R. Velasco-Fernández, S.G.F. Bukkens, Report on the experience of applications of the Nexus Structuring Space in Quantitative Story-Telling. MAGIC (H2020–GA 689669) Project Deliverable 4.4, (2020). http://m agic-nexus.eu/documents/deliverable-44-report-nexus-structuring-space.
- [91] EUROSTAT, Complete energy balances [nrg_bal_c], (2019).
- [92] EUROSTAT, Export of biofuels by partner country. [nrg_te_bio], (2019).
- [93] EUROSTAT, Population change Demographic balance and crude rates at national level. [demo_gind], (2019).
- [94] FAO, FAOSTAT Database, (2019).
- [95] CBS-PBL-RIVM-WUR, Use of renewable energy for transport, 2005-2017 (indicator 0535, versie 18, 11 April 2019). Environmental Data Compendium, (2019). https://www.clo.nl/en/indicators/en0535-biofuels (accessed June 11, 2019).
- [96] Tweede Kamer der Staten-Generaal Nederland, Rapportage Energie voor Vervoer in Nederland 2017 (bijlage bij 30196,nr.605), (2018) 1–58. https://www. rijksoverheid.nl/documenten/rapporten/2018/07/03/bijlage-1-rapportage-ener gie-voor-vervoer-in-nederland-2017.
- [97] E. van Thuijl, E.P. Deurwaarder, European Biofuel Policies in Retrospect. Report ECN-C-06-016, May 2016, (2016) 50. https://publications.ecn.nl/ECN-C-06-016.
- [98] CBS, Hernieuwbare energie in Nederland 2018, (2019). https://longreads.cbs. nl/hernieuwbare-energie-in-nederland-2018/.
- [99] EUROSTAT, Renewable energy statistics, (2020). https://ec.europa.eu/eurostat /statistics-explained/index.php/Renewable_energy_statistics#of renewable_ener gy_used_in_transport_activities_in_2018 (accessed July 1, 2020).
- [100] C. Sheng Goh, T. Mai-Moulin, M. Junginger, Sustainable biomass and bioenergy in the Netherlands: Report 2015, Utrecht, the Netherlands, 2016. https://english. rvo.nl/sites/default/files/2016/05/Sustainable-biomass-bioenergy-netherlands. pdf.
- [101] Government of the Netherlands, Besluit hernieuwbare energie vervoer 2015, (2015). https://wetten.overheid.nl/BWBR0035839/2015-01-01.
- [102] F. Hillairet, V. Allemandou, K. Golab, Analysis of the current development of household UCO collection systems in the EU. GREENEA Technical Report, 23 May 2016., Coivert, France, 2016. https://theicct. org/sites/default/files/publications/Greenea Report Household UCO Collection in the EU ICCT_20160629.pdf.
- [103] T. Tsoutsos, T. Stavroula, Assessment of best practices in UCO processing and biodiesel distribution. D4.3 - Guide on UCO processing and biodiesel distribution methods. Version 3, Crete, 2013. https://ec.europa.eu/energy/intelligent/ projects/sites/iee-projects/files/projects/documents/d4.3_guide_on_uco_ processing_and_biodiesel_distribution_v4tuc.pdf.
- [104] CNMC, Estadística de biocarburantes, Madrid, 2019.
- [105] J. Giuntoli, Advanced Biofuel Policies in select EU Member States: 2018 Updated, 2018.
- [106] European Court of Auditors, The EU system for the certification of sustainable biofuels, Publications Office of the European Union, Luxembourg, 2016. https://doi.org/10.2865/82411.
- [107] Toop, G., S. Alberici, M. Spoettle, H. van Steen, U. Weddige, Trends in the UCO market, London, 2014. https://assets.publishing.service.gov.uk/government/upl oads/system/uploads/attachment_data/file/307119/trends-uco-market.pdf.
- [108] Nederlandse Emissieautoriteit (NEa), Rapportage Energie voor Vervoer 2018. Naleving verplichtingen wet- en regelgeving Energie voor Vervoer, 2019. https://www.rijksoverheid.nl/documenten/rapporten/2019/07/04/rapportage -energie-voor-vervoer-in-nederland-2018.
- [109] G. Reijn, Fraude met eenderde van alle biodiesel, Kampens bedrijf zou 'miljoenen' hebben verdiend, Volkskrant, 23 May 2019. (2019). https://www.volkskrant. nl/nieuws-achtergrond/fraude-met-eenderde-van-alle-biodiesel-kampens-bedr ijf-zou-miljoenen-hebben-verdiend~b01c604b/?referer=https%3A%2F% 2Pwww.google.com%2F.
- [110] Nederlandse Emissieautoriteit (NEa), Fraude met biodiesel: aanpakken en investeren in vertrouwen (in Dutch), NEA Nieuws, 06 Sept. 2019. (2019).

https://www.emissieautoriteit.nl/actueel/nieuws/2019/09/06/fraude-metbiodiesel.

- [111] B. Holmatov, A.Y. Hoekstra, M.S. Krol, Land, water and carbon footprints of circular bioenergy production systems, Renew. Sustain. Energy Rev. 111 (2019) 224–235, https://doi.org/10.1016/j.rser.2019.04.085.
- [112] T. Bole, M. Londo, J. van Stralen, A. Uslu, Overcoming the initial investment hurdle for advanced biofuels: An analysis of biofuel-related risks and their impact on project financing., Amsterdam, 2010. https://ec.europa.eu/energy/intelligent /projects/sites/iee-projects/files/projects/documents/elobio_biofuel_related_ri sks.pdf.
- [113] S. Rayner, Uncomfortable knowledge: The social construction of ignorance in science and environmental policy discourses, Econ. Soc. 41 (2012) 107–125, https://doi.org/10.1080/03085147.2011.637335.
- [114] S. Morgan, Dutch do Danish deal to hit clean power target. EURACTIV, 22 June 2020., (2020). https://www.euractiv.com/section/energy/news/dutch-dodanish-deal-to-hit-clean-power-target/ (accessed June 24, 2020).
- [115] S. Michalopoulos, Industry source: one third of used cooking oil in Europe is fraudulent, EURACTIV. (2019).
- [116] S. Michalopoulos, EU lacks 'complete overview' about used cooking oil origins. EURACTIV, 24 June 2020, (2020). https://www.euractiv.com/section/agricultu re-food/news/eu-lacks-complete-overview-about-used-cooking-oil-origins/?ut m_source=EURACTIV&utm_campaign=5b42d90d68-AgriFood Brie f_COPY_02&utm_medium=email&utm_term=0_c59e2fd7a9-5b42d90d68-114980147 (accessed June 25, 2020).
- [117] R. Velasco-Fernández, T. Dunlop, M. Giampietro, Fallacies of energy efficiency indicators: recognizing the complexity of the metabolic pattern of the economy, Energy Policy 137 (2020), 111089, https://doi.org/10.1016/j. enpol.2019.11089.
- [118] T. Völker, Z. Kovacic, R. Strand, Indicator development as a site of collective imagination? The case of European Commission policies on the circular economy, Cult. Organ. 26 (2020) 103–120, https://doi.org/10.1080/ 14759551.2019.1699092.
- [119] T. Voelker, K. Blackstock, Z. Kovacic, J. Sindt, R. Strand, K. Waylen, The role of metrics in the governance of the water-energy-food nexus within the European Commission, J. Rural Stud. (2019), https://doi.org/10.1016/j. jrurstud.2019.08.001.
- [120] D. Waltner-Toews, A. Biggeri, B. de Marchi, S. Funtowicz, M. Giampietro, M. O'Connor, Post-normal pandemics: Why COVID-19 requires a new approach to science. STEPS Centre blog post, 25 March 2020., 2020. (n.d.). https://steps-ce ntre.org/blog/postnormal-pandemics-why-covid-19-requires-a-new-approach-to -science/ (accessed June 21, 2020).
- [121] J.Z. Muller, The Tyranny of Metrics | Princeton University Press, Pinceton University Press, Princeton, NJ, 2018. https://press.princeton.edu/books/h ardcover/9780691174952/the-tyranny-of-metrics (accessed June 21, 2020).
- [122] A. Renner, J.J. Cadillo-Benalcazar, L. Benini, M. Giampietro, Environmental pressure of the European agricultural system: Anticipating the biophysical consequences of internalization, Ecosyst. Serv. accepted f (2020).
- [123] IPBES, Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services., IPBES secretariat, Bonn, Germany, 2019. https://doi.org/10.5281/zenodo.3553579.

- [124] IPCC, Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems., 2019. https://www.ipcc.ch/ srccl/ (accessed September 19, 2020).
- [125] FAO, The state of food security and nutrition in the world 2019 Safeguarding against economic slowdowns and downturns, 2019. http://www.fao.org/3/ca5 162en/ca5162en.pdf.
- [126] Ecofys, Technical assistance in realisation of the 2018 report on biofuels sustainability. Biofuels, biomass & biogas used for renewable energy generation. Final report., (2019). https://ec.europa.eu/energy/sites/ener/files/documents/te chnical assistance_in_realisation_of_the_2018_report_on_biomass_sustainabilit y-final_report.pdf.
- [127] H. Valin, D. Peters, M. van den Berg, S. Frank, P. Havlik, N. Forsell, C. Hamelinck, The land use change impact of biofuels consumed in the EU. Quantification of area and greenhouse gas impacts. Project number: BIENL13120, Utrecht, the Netherlands, 2015. https://ec.europa.eu/energy/sites/ener/files/doc uments/Final Report_GLOBIOM_publication.pdf (accessed September 21, 2020).
- [128] J.J. Cadillo-Benalcazar, A. Renner, M. Giampietro, A multiscale integrated analysis of the factors characterizing the sustainability of food systems in Europe, J. Environ. Manage. 271 (2020), 110944, https://doi.org/10.1016/j. jenvman.2020.110944.
- [129] M. Padella, A. O'Connell, M. Prussi, What is still limiting the deployment of cellulosic ethanol? Analysis of the current status of the sector, Appl. Sci. 9 (2019) 4523, https://doi.org/10.3390/app9214523.
- [130] RECOIL, Promotion of used cooking oil recycling for sustainable biodiesel production (RECOIL), EC Intell. Energy Eur. Proj. Database. (2015). https://ec. europa.eu/energy/intelligent/projects/en/projects/recoil (accessed August 25, 2020).
- [131] G. Philippidis, H. Bartelings, J. Helming, R. M'barek, E. Smeets, H. van Meijl, Levelling the playing field for EU biomass usage, Econ. Syst. Res. 31 (2019) 158–177. https://doi.org/10.1080/09535314.2018.1564020.
- [132] European Commission, Communication from The Commission to The European Parliament, The Council, The European Economic and Social Committee and The Committee of the Regions: Innovating for Sustainable Growth: A Bioeconomy for Europe {SWD(2012) 11 final}. COM(2012) 60 final., (2012). http://ec.europa. eu/research/bioeconomy/pdf/official-strategy en.pdf (accessed March 8, 2020).
- [133] N. Pavlenko, S. Searle, C. Baldino, Assessing the potential advanced alternative fuel volumes in Germany in 2030, 2019. https://theicct.org/sites/default/files /publications/Advanced_fuels_potential_Germany_20190916.pdf (accessed April 3, 2020).
- [134] J.J. Cheng, G.R. Timilsina, Status and barriers of advanced biofuel technologies: a review, Renew. Energy. 36 (2011) 3541–3549, https://doi.org/10.1016/j. renene.2011.04.031.
- [135] S. Michalopoulos, Netherlands mulls end to used cooking oil double-counting, EURACTIV. (2019).
- [136] G. Lakoff, Why it matters how we frame the environment, Environ. Commun. 4 (2010) 70–81, https://doi.org/10.1080/17524030903529749.
- [137] J.R. Ravetz, Usable knowledge, usable ignorance, Knowledge 9 (1987) 87–116, https://doi.org/10.1177/107554708700900104.